Author’s response to reviews

Title: Health care restructuring and family physician care for those who died of cancer

Authors:

Frederick I. Burge (fred.burge@dal.ca)
Beverley Lawson (bev.lawson@dal.ca)
Grace Johnston (grace.johnston@dal.ca)
Gordon Flowerdew (gordon.flowerdew@dal.ca)

Version: 3 Date: 24 December 2004

Author’s response to reviews:

Only formatting changes have been made:

- Reference citations in the manuscript text are now before the full stop

- The project support acknowledgement has been moved to the Acknowledgements section

- All references contain all authors' names and no usage of "et al"

- The table legend is only on the same page as the table, directly above the table

- The figures have been cropped and these files don't include titles or figure numbers